PREVALENCE AND TRANSMISSION OF ASYMPTOMATIC C. DIFFICILE

EVALUATION OF THE PREVALENCE AND TRANSMISSION OF ASYMPTOMATIC CLOSTRIDIOIDES DIFFICILE CARRIAGE IN THE HAMILTON IN-PATIENT SETTING USING MULTI-LEVEL MODELLING

By SYDNEY GEORGE, B.Sc.

A Thesis Submitted to the School of Graduate Studies In Partial Fulfilment of the Requirements for the Degree Master of Science In Health Research Methodology

McMaster University © Copyright by Sydney George, September 2020

MASTER OF SCIENCE (2020)

Health Research Methodology

McMaster University Hamilton, Ontario, Canada

TITLE:Evaluation of the Prevalence and Transmission of
Asymptomatic Clostridioides difficile Carriage in the
Hamilton In-Patient Setting Using Multi-Level Modelling

AUTHOR: Sydney George, B.Sc.

SUPERVISOR: Dr. Dominik Mertz, MD, MSc, FMH

NUMBER OF PAGES: xi,71

LAY ABSTRACT

C. difficile infection (CDI) is a severe infectious disease. Patients with asymptomatic *C. difficile* present a risk to others as they can contribute to the spread and development of hospital-associated CDI. We are currently unsure of the proportion of adult in-patients colonized with asymptomatic *C. difficile*. Identifying these carriers early on in their hospital stay is imperative to reduce CDI rates in health care settings. Our study objectives were to determine the best screening strategies to identify asymptomatic carriers, identify risk factors for carriage, and understand the transition from asymptomatic *C. difficile* to symptomatic CDI. We demonstrated that being female, being recently hospitalized or previously having CDI may increase a patient's risk of being an asymptomatic carrier. Also, timely screening throughout a hospital stay in addition to admission screening helped identify more colonized in-patients. Lastly, we determined that 1 in 5 carriers would go on to develop symptomatic CDI infection.

ABSTRACT

Background: *C. difficile* is one of the primary infectious causes of morbidity and mortality in Canada. Colonized patients can pose a risk to others as a factor in the transmission and development of hospital-associated *C. difficile* infections. Despite immense efforts and resources invested in the reduction in *C. difficile* transmission within Canada and Hamilton Health Sciences – further reduction in these rates are unlikely, and novel screening strategies are imperative in this field of study.

Methods: This project was a retrospective cohort study of adult in-patients admitted to either The Juravinski, Hamilton General, or St. Joseph's Healthcare Hamilton Hospitals from January to April 2018 and September 2018 to August 2019. MSRA/VRE swabs were collected during admission or through universal point prevalence screening and subsequently tested for colonization.

Results: From the 1056 patients in the data sample, 72 were colonized with asymptomatic *C. difficile* resulting in a prevalence rate of 6.81%. In-patient point prevalence screening strategies identified more carriers than admission swabs alone (p < 0.001). Risk factors for colonization on admission were being female (OR 2.66, 95% CI 1.02-8.33) and previous CDI (OR 4.76, 95% CI 1.49 – 13.86). During hospitalization, risk factors for colonization were previous CDI (OR 4.75 95% CI 2.14-9.94) and recent hospitalization within the last 12 months (OR 2.35, 95% CI 1.30-4.42). The multi-level Cox PH model identified those with a recent hospitalization (OR 2.21, 95% CI 1.32 – 3.73) and those with previous CDI (OR 2.40, 1.34 – 4.30) were twice as likely to develop asymptomatic C. difficile colonization throughout hospitalization.

Conclusion: The addition of universal point prevalence screening in addition to admission screening helped identify more than double the amount of carriers in the population. Moreover, a previous hospitalization, previous CDI, and being female may indicate patients at the highest risk of colonization.

ACKNOWLEDGEMENTS

I would like to thank my supervisor, Dr. Dominik Mertz, for his constant support, guidance, and mentorship throughout my Health Research Methodology studies. Secondly, I would like to thank my committee members Drs. Marek Smieja, Robby Niewulaat, and Allison McGeer for their feedback and direction throughout the completion of my thesis project. I would also like to thank Julia Maciejewski for processing the laboratory data needed for my thesis. Lastly, I would like to thank my family and friends for their constant and unwavering support over the years towards my academic pursuits.

TABLE OF CONTENTS

Lay Abstract		
Abstract		
Acknowledgements		
Table of Contents	vi	
List of Figures and Tables	viii	
List of Abbreviations and Symbols	x	
Declaration of Academic Achievement	xi	
PART I: INTRODUCTION & NARRATIVE REVIEW	1	
An Introduction to Clostridioides Difficile	1	
Transmission of CDI	2	
Epidemiology & Disease Burden of <i>C. difficile</i> in the Canadian Context	4	
Asymptomatic C. difficile Colonization	6	
Epidemiology of Asymptomatic <i>C. difficile</i> Colonization	7	
Risk Factors for <i>C. difficile</i> Colonization	8	
Laboratory Detection of <i>C. difficile</i>	8	
Testing Strategies for <i>C. difficile</i> Colonization	10	
Asymptomatic C. difficile colonization progression to Symptomatic CDI	11	
PART II: RETROSPECTIVE COHORT STUDY	14	
Objectives	14	
Investigation Plan	16	
Study Procedures	21	

Analysis	24
PART III: RESULTS	30
Descriptive Results	30
Comparison of Testing Strategies	32
Risk Factors for Colonization, Upon Admission	34
Risk Factors for Colonization, Point Prevalence	35
Time-to-Colonization of <i>C. difficile</i>	36
Progression to CDI	38
PART IV: DISCUSSION	54
Summary of Findings	54
Limitations	60
PART V: CONCLUSION AND FUTURE STEPS	65
APPENDIX	65
REFERENCE	

LIST OF FIGURES AND TABLES

<u>Tables</u>

Table 1:	Baseline Characteristics of Adult Inpatients Admitted to The Juravinski, Hamilton General or St Joseph's Healthcare Hamilton Hospitals	40
Table 2a:	Comparison of <i>C. difficile</i> Screening Identification Methods of High-Risk MRSA/VRE In-patients	42
Table 2b:	Comparison of <i>C. difficile</i> Screening Identification Methods of High-Risk MRSA/VRE In-patients	43
Table 3:	Comparison of Patients who are Positive at Admission and Positive during Hospital Stay	44
Table 4:	Predictors of <i>C. difficile</i> Colonization and Model Fit of Multivariable Analysis (Admission)	46
Table 5:	Predictors of <i>C. difficile</i> Colonization and Model Fit of Multivariable Analysis (Point Prevalence)	47
Table 6:	Results of Cox mixed-effects model on time-to- colonization of Asymptomatic <i>C. difficile</i> Colonization	48
Table 7:	Comparison of <i>C. difficile</i> Carriage Progression to Symptomatic CDI	49
Figures		

Figures

Figure 1:	Retrospective Cohort Study, in-Patient progression	51
Figure 2:	Kaplan Meier Estimates for Time to Colonization during Hospitalization of Asymptomatic <i>C. difficile</i> stratified by Recent Hospitalization	52
Figure 3:	Kaplan Meier Estimates for Time to Colonization during Hospitalization of Asymptomatic <i>C. difficile</i> stratified by Ward	53

Figure 4:Kaplan Meier Estimates for Time to Colonization during
Hospitalization of Asymptomatic C. difficile stratified by
Age Categories younger and older than 65 years54

LIST OF ALL ABBREVIATIONS AND SYMBOLS

CA-CDI	Community Associated Clostridioides difficile Infection
CCCNA	Cell Culture Cytotoxicity Neutralization Assay
CDC	Centers for Disease Control and Prevention
CDI	Clostridioides difficile Infection
EIA	Enzyme Immunoassay
GEE	Generalized Estimating Equation
HA-CDI	Hospital Associated Clostridioides difficile Infection
HHS	Hamilton Health Sciences
HRLMP	Hamilton Regional Laboratory Medicine Program
LAMP	Loop-Mediated Isothermal Amplification
MRSA	Methicillin-Resistant Staphylococcus aureus
NAP1	Nucleosome Assembly Protein-1
PCR	Polymerase Chain Reaction
VRE	Vancomycin-Resistant Enterococci

DECLARATION OF ACADEMIC ACHIEVEMENT

I, Sydney George, declare my thesis to be my own research work. I am the sole author of this thesis document and was involved in all stages of the research project under the supervision of Dr. Dominik Mertz. The following individuals contributed to the editing and refinement of my thesis work and acting as the members of my thesis committee: Dr. Dominik Mertz, Dr. Marek Smieja, Dr. Robby Nieuwlaat and Dr. Allison McGeer. Julia Maciejewski processed laboratory results for data collection. To my knowledge, the content of this document does not infringe on any copyrights.

PART I: INTRODUCTION & NARRATIVE REVIEW

An Introduction to Clostridioides Difficile

Clostridioides difficile (*C. difficile*) is a gram-positive bacterium and a spore-forming obligate anaerobe first described as *Bacillus difficilis* by Hall and O'Toole in 1935(1). Up until the early 2000s, infection by *C. difficile* was not viewed as a threat to public health and a treatable complication of antimicrobial therapy(2). However, *C. difficile* infection (CDI) has re-emerged as a severe infectious disease with increased complications(2). Currently, CDI is the leading cause of antibiotic-associated diarrhea in Europe and North America and affects more than 300,000 hospitalized patients yearly in the United States(2). Moreover, there were 20, 623 cases of health care associated CDI from 2009-2015 in Canada(3). *C. difficile* became a notifiable disease under national surveillance in 2009 through the Canadian Notifiable Disease Surveillance System (CNDSS)(4).

C. difficile is commonly found in water, air, and soil – as well as human feces and hospital surfaces(5). Spores of *C. difficile* are dormant cells and are resistant to high temperatures, ultraviolet light, harsh chemicals, disinfectants, and antibiotics(6). *C. difficile* bacterium is known to colonize the colon primarily due to the large bowel being an anaerobic environment, and antibiotic medication can eliminate the absence of competitor flora(7). Following or during colonization, vegetative *C. difficile* bacterium releases toxin A (TcdA) and toxin B (TcdB)(8).

TcdA acts primarily on the intestinal epithelium, causing fluid secretion, general inflammation, and necrosis of tissue, whereas TcdB acts as a potent cytotoxin(8). The clinical manifestations of *C. difficile* infection can range from mild diarrhea to life-threatening toxic megacolon or pseudomembranous colitis(9). The severity of CDI depends on the virulence (e.g. NAP1) of the pathogen and host characteristics(10). *C. difficile* is a common nosocomial pathogen, with reservoirs regularly being patients and contaminated environments(7).

In disease-endemic settings, the prevalence of colonized but asymptomatic hosts is typically higher than symptomatic CDI cases(11). Therefore, asymptomatic *C. difficile* colonized patients may be an essential infection reservoir and a source of transmission(11). Prevalence of *C. difficile* colonization differs between populations, depending on several patient, environmental, and pathogen characteristics(11). Nonetheless, it is imperative to understand vehicles of transmission for *C. difficile* in healthcare settings – which remains an ongoing challenge to the world of infection prevention and control.

Transmission of CDI

C. difficile is transmitted via the oral-fecal route(5). Spores are the primary infectious vehicle of *C. difficile*, as vegetative *C. difficile* bacterium is unlikely to

survive in an oxygenated environment and outside of the acidic environment of the stomach or the host in general(12).

CDI is differentiated between hospital-acquired and community acquired, where;

- a) Hospital-Acquired CDI (HA-CDI): The symptoms of CDI were not present on or within 72 hours of admission, or CDI symptoms are present at the time of admission however are related to a previous admission to a healthcare facility within the last four weeks (13).
- b) Community-Acquired CDI (CA-CDI): The symptoms of CDI were present on admission or developed within < 72 hours after admission, with no previous hospital admission within the last four weeks (13).

C. difficile has primarily been considered a hospital-acquired infection, and the Centers for Disease Control and Prevention (CDC) has estimated that 94% of all CDI is related to healthcare setting exposure(14). A population-based study, however, found that a substantial proportion of CDI was community-acquired, and other surveillance studies have found the same percentage of communityacquired CDI cases among all CDI cases(15-17). Moreover, community-acquired CDI affects previously considered low-risk populations such as younger patients with no antibiotic exposure and recent hospitalization(18). A study by Kuntz et al

identified that 17% of CA-CDI patients did not have any of the traditional risk factors for CDI (19).

Furthermore, due to *C. difficile* spores' antibiotic resistance, they can remain in the gastrointestinal tract, making recurrent disease following treatment likely(20). Recurrence of symptoms after effective treatment of CDI is widespread, and historically recurrent CDI occurs in 20-30% of patients(20). This rate of recurrence may be higher since the appearance of the virulent NAP1 epidemic strain(21).

Epidemiology & Disease Burden of C. difficile in the Canadian Context

C. difficile is one of the primary infectious causes of morbidity and mortality in Canada and is the most frequent cause of infectious diarrhea in hospitals and long-term care facilities(22). Between 2005-2015 alone, there were approximately 40,000 new cases of *C. difficile* infection in Ontario – half of which were acute care hospital-associated(23). Since 2002, there has been an increase in CDI incidence and severity in North America(24). A Quebec study reported that between 1991 and 2003, there was a 330% increase in CDI cases per 100,000 people(23). In 2011, the incidence rate of CDI among Canadian provinces was 535 per 100,000 admissions (23). MSc Thesis – S. George; McMaster University – Health Research Methodology.

Moreover, 30-day mortality and the proportion of severe complications (development of megacolon, colectomy, for example) increased between 1991 and 2003(2). *C. difficile* disproportionately affects the elderly, with 93% of CDI deaths occurring in persons older than 65 years(2). The disproportionate prevalence of cases in the elderly population is of concern with a current rising average age of Canadians.

The incidence of community-acquired CDI cases has increased as well. Khanna et. al demonstrated that community-acquired CDI raised 5.3 fold in 10 years in Olmsted County, Minnesota(25). Community-acquired CDI also tended to affect a different population in comparison to hospital-acquired CDI, with most cases being: younger (median age 50), likely to be female and with fewer comorbidities than patients with hospital-acquired cases(2). A study by Reveles et. al found an increase of community associated CDI from 8.3% to 26.7% of patients between 2002 to 2014(26). Patient characteristics of communityacquired cases were: less likely to have a severe infection, less likely to be on antibiotic treatments, and less likely to have cancer than hospital-acquired CDI patients(2).

Researchers have begun to explore why there is an emergence of virulent CDI, and it is thought to be attributed to hypervirulent strains(27). It has been reported in the literature that the ribotype NAP1/B1/027 strain of CDI may be

more virulent compared to other pathogenic isolates, and ribotype 078, 126, 056, and 018 may contribute to more severe CDI outcomes as well(2).

CDI has also been associated with substantial costs to the Canadian health care system. A study in Ontario has shown that acquiring a *C. difficile* infection increases the median length of an in-patient stay by six days(28). The mean attributable costs of a CDI infection per hospitalized patient ranged from CAD \$ 10,961 to CAD \$ 36,960(29). Another study estimated that mean attributable 1-year costs were \$13,217 (95%CI, \$12,062-\$14,388) unadjusted for survival and \$10,700 (95%CI, \$9,811-\$11,645) adjusted for survival(29). A Canadian model estimated a combined 37,900 cases of CDI in the hospital and the community in 2012, had a societal cost of \$ 281 million, of which 92% were in-hospital costs(30). The re-emergence of CDI affects a vulnerable sector of the population and has resulted in more frequent surgical complications, severe outcomes of disease, increased health care costs, increased mortality rates, and will be a significant contributor to economic burden in the years to come without interventions.

Asymptomatic C. difficile Colonization

Asymptomatic *C. difficile* colonization is defined as a condition where a patient has detectable concentrations of *C. difficile* or its toxin gene in the

absence of symptoms of CDI (diarrhea, and colonoscopic or histopathologic findings consistent with pseudomembranous colitis)(11). Asymptomatic *C. difficile* colonized patients may be important reservoirs of infection. Colonized patients can present a risk to others as a factor in the transmission and development of hospital-associated *C. difficile* infections, especially with emergent hypervirulent strains being of concern. 29% of isolates from hospital associated CDI were related to isolates obtained from colonized patients (31). More than 30 years ago, 79% of hospital-associated CDI were reported to be linked to asymptomatic carriers(32). Literature also suggests that asymptomatic *C. difficile* colonization with non-toxigenic strains may be protective of disease progression to symptomatic CDI due to a coordinated immune response to *Clostridioides* toxins(11, 33). Nonetheless, it is essential to establish the contribution of asymptomatic patients as vehicles of disease transmissions

After exposure to *C. difficile*, an estimated two-thirds of patients who acquire the organism will become asymptomatic carriers(34). The terms asymptomatic carriage or asymptomatic colonization are used interchangeably within the literature, with the term carriage used more often to refer to persistent colonization(11).

Epidemiology of Asymptomatic C. difficile Colonization

Asymptomatic C. difficile colonization is more common than symptomatic CDI(11). However, the prevalence of colonization depends significantly on the patient population and environment – with considerably different prevalence rates between groups. Due to the nature of asymptomatic carriage, it is difficult to understand its clinical implications fully and testing patient populations for colonization is required. The highest colonization rates are in infants during their first year of life – who are commonly colonized and have a range of carrier rates being between 18-90%(11). Pediatric oncology in-patients have been found to have prevalence rates of colonization approximating 25% (35). Moreover, cystic fibrosis patients have colonization rates between 18-47%(11). Healthy adults in the general population have been found to have prevalence rates from 0-15%(11). A large Canadian study found that in a mixed population of hospital admissions, colonization rates were at 4.05%, and other literature has suggested the in-patient population prevalence rate to be between 4-29%(11, 36). The significant heterogeneity in presented prevalence rates is probably due to differing populations, screening strategies, environment and clinical settings, and detection methods.

Risk Factors for C. difficile Colonization

Risk factors for *C. difficile* colonization in adult in-patient populations have been identified in recently published literature. Risk factors for *C. difficile* colonization include a previous hospitalization within the last 12 months, medication therapies including proton pump inhibitors, H2 blockers, chemotherapy, cephalosporin use during admission and presence of toxin B antibodies – which have also been suggested to be protective against the development of CDI(11, 36).

Prevalence within the healthy adult population remains considerably low. However, the prevalence increases significantly once a patient is within the healthcare system. For example, a study found that long-term care homes have a rate of colonization between 0-51%, and a high prevalence of asymptomatic *C*. *difficile* carriage is common among healthcare workers, patients in rehabilitation centres or hospitals in-patients(11). A study by Kong et al. reported that a previous history of CDI and the use of corticosteroids are risk factors of *C. difficile* colonization(36). Lastly, patients with multiple comorbidities have been associated with higher *C. difficile* colonization as well(37). Understanding the landscape of potential risk factors may help guide screening strategies for *C. difficile* colonization detection.

Laboratory Detection of C. difficile

The process of laboratory detection for C. difficile involves detecting toxins A and B of toxigenic C. difficile in the stool(38). There are a variety of diagnostic methods available, each with their own respective advantages and disadvantages. Currently, the gold standard for detecting C. difficile toxin is cell culture cytotoxicity neutralization assay (CCCNA) and anaerobic toxigenic culture (TC)(38). Enzyme immunoassay (EIA) is another widely used detection method. The advantages of the tests are that CCCNA has high sensitivity, and EIA has a rapid turnaround, has an associated low-cost and is a simple technique(38). However, CCCNA is time-consuming, TC is slow with a 48-72 hour turnaround, and EIA has low sensitivity with ranges of 50-90% (39). PCR is another commonly used tool for detecting C. difficile toxins, with rapid turn around and high sensitivity, however, with PCR needing more specific equipment - the use of PCR as a primary detection method for *C. difficile* would be impractical for many clinical laboratories(38). Loop-mediated isothermal amplification (LAMP) is recently another diagnostic tool used for C. difficile detection, which has the benefit of being both sensitive (98%) and specific (98%) with a rapid one-hour turnaround time and is comparable to both PCR and EIA methods(40). Literature, however, debates on the importance of which toxin LAMP targets. LAMP targeting toxin A may not be ideal as toxin B is proven to be more virulent and there is the existence of toxin A-negative C. difficile strains, which studies have shown to make up 21.4% of CDI strains and may also present a challenge (41).

Nonetheless, LAMP may be a cost-effective, rapid and reliable diagnostic method to use for *C. difficile* toxin detection.

Testing Strategies for C. difficile Colonization

Although the role of asymptomatic *C. difficile* carriers as vehicles for transmission is poorly understood, literature using whole genome-wide sequencing and molecular typing suggests they may play an essential role in infection spread. Preliminary research has suggested that early identification and intervention of asymptomatic colonized patients with *C. difficile* may reduce transmission. Longtin et. al 2016 demonstrated that universal admission screening for *C. difficile* carriage and isolation in conjunction with contact precautions of identified carriers reduced hospital CDI by 62%, and estimated that 60 cases of acute CDI were prevented(42). In consideration, however, mass screening techniques of all newly admitted patients for *C. difficile* can be time-consuming and costly, while putting an additional workload strain on hospital microbiology labs.

Patients high-risk for MRSA/VRE carriage have previous hospitalization as a risk factor, and systematic screening of this population is recommended. Similarly, previous hospitalization within three months is a risk factor for *C*.

difficile carriage(11). Therefore, one could conclude that patients who are high risk for MRSA/VRE carriage may be high-risk for *C. difficile* colonization, and repurposing MRSA/VRE swabs may be a method of screening patients that is well integrated into existing processes. A study with 31 participants determined that a screening strategy based on high-risk MRSA/VRE carriers could identify 74% of all *C. difficile* carriers(43). Another study revealed that 9.5% of inpatients were carriers upon admission; however, routine weekly screening identified another 7.2% of colonized patients – who acquired *C. difficile* colonization after hospital admission(33). Another study had reported an increase of 50% in colonization rates throughout a hospital stay, despite a relatively low colonization rate amongst admitted patients (2.1%)(32). Therefore it is imperative that both admission and point prevalence screenings are taken into account for developing a screening method for asymptomatic *C. difficile* carriage detection in order to capture all colonized patients.

Asymptomatic C. difficile colonization progression to Symptomatic CDI

While the disease process of CDI is well understood, the role of asymptomatic *C. difficile* colonization and its development into symptomatic CDI has a weaker understanding. The risk of asymptomatic carriage and its progression to symptomatic CDI is highly variable, with heterogeneity relating to patient characteristics, bacterial factors and other extrinsic risk factors(44).

Patients who had acquired asymptomatic C. difficile colonization were less frequently colonized with the hypervirulent NAP1 strain, in comparison to those with symptomatic CDI infection – and may suggest strain may influence whether a patient is asymptomatic or symptomatic (11). Moreover, in a published systematic review, 21.8% of adult patients who were colonized upon admission developed CDI during their hospital stay – resulting in patients colonized having a higher risk of symptom development than non-colonized patients(45). Another study reported a rate of 2.1% of CDI development from colonization in a low-risk mixed patient population. Rates of CDI development from colonization may be much higher in populations such as hematopoietic stem cell transplant patients; a study found that 85.5% developed CDI in comparison to 17.2% of non-carriers (odds ratio (OR) 68.5, 95% CI 11.4-416.2) (46). Asymptomatic colonization has also potentially been considered a protective factor against CDI. A meta-analysis of 4 studies demonstrated overall that patients colonized with C. difficile had a lower risk of subsequently developing CDI(47). Furthermore, the evidence shows that patients colonized specifically by only non-toxigenic strains are protected from progressing to CDI, while those colonized with toxigenic C, difficile at admission progress to CDI more frequently (RR 5.86, 95% CI 4.21-8.16)(44, 47). Our goal is to assess the natural history of C. difficile carriage in in-patient populations in order to inform both future interventional studies and screening programs.

PART II: RETROSPECTIVE COHORT STUDY

Materials and Methods

INTRODUCTION

The estimation of the prevalence of asymptomatic *C. difficile* colonization in the adult in-patient population is relatively uncertain. Moreover, in order to reduce the rates of transmission of *C. difficile* within hospital settings, the identification of carriers early on in the in-patient process is imperative. Therefore a longitudinal retrospective cohort study was established to: 1) identify optimal screening strategies for asymptomatic carriers, 2) understand the risk factors for asymptomatic *C. difficile* carriage at admission and during a hospital stay, and 3) understand the transition from asymptomatic *C. difficile* carriage to symptomatic CDI as well as the natural history of the disease.

OBJECTIVES

Subproject 1: Screening, Admission & Point Prevalence

Primary Objective

MSc Thesis – S. George; McMaster University – Health Research Methodology.

To determine whether admission screening swabs of high-risk individuals for MRSA/VRE (done as part of routine MRSA/VRE surveillance) can identify the majority (>80%) of in-patients who were colonized with *C. difficile* before hospital entry, using routine universal point prevalence testing of the entire unit as the reference standard.

Secondary Objective

Estimate the incidence of *C. difficile* colonization acquisition during hospital stays on high-risk wards by documenting the proportion of individuals negative at admission who are later found to be positive for asymptomatic colonization.

Sub-Project 2: Symptomatic CDI & Transmission

Primary Objective

To determine the incidence of symptomatic *C. difficile* infection (CDI) amongst colonized patients.

INVESTIGATION PLAN

Summary of Study Design

This project was a retrospective cohort study of adult in-patients admitted to one of the study wards over eight months. High-risk in-patients routinely screened for MRSA/VRE upon admission or throughout routine, point-prevalence screening had their samples tested for *C. difficile* colonization. Amongst patients tested for *C. difficile*, a retrospective chart review was conducted to identify colonized in-patients and potential risk factors of interest for asymptomatic colonization and the development of symptomatic *C. difficile* infection.

Retrospective Cohort Study

This project was a retrospective cohort study of adult in-patients admitted to either The Juravinski, Hamilton General, or St. Joseph's Healthcare Hamilton Hospitals from January to April 2018, and September 2018 to August 2019. Highrisk in-patients routinely screened for MRSA/VRE upon admission or throughout routine, point-prevalence screening had their swab samples tested for *C. difficile*. Among patients tested for *C. difficile*, a retrospective chart review was conducted to identify colonized in-patients and potential risk factors of interest for asymptomatic colonization and the development of symptomatic *C. difficile* infection.

Rationale for Study Design

In an observational cohort study, participants are selected based on exposure status and do not have the outcome of interest at baseline(48). Participants are grouped into either exposed or not exposed at study initiation and are followed-up over a specified time for assessment of the occurrence of the outcome of interest(48). In a retrospective cohort study, outcomes of interest have occurred in the past, and data is collected from databases or health care records. A cohort study is an advantageous design for assessing causal relationships in comparison to other observational study designs, such as casecontrol studies, due to the temporal sequence of risk factors recorded before the occurrence of the outcome.

Moreover, a cohort study allows researchers to examine multiple outcomes for a given exposure and determine the incidence and prevalence rates of disease. In particular, retrospective study designs are more cost-efficient and shorter than prospective study designs. Limitations of a retrospective cohort study typically include limited control the investigator has over data collection, resulting in sometimes inaccurate or incomplete study data. A cohort study design allows for the evaluation of the natural history of colonization and calculation of incidence and prevalence of both asymptomatic colonization and symptomatic *C. difficile* infection.

Also, given the exploratory nature of the project, a cohort study allows for examining several risk factors of interest (e.g., antibiotic exposure, duration of hospital stay, comorbidities of interest) for asymptomatic *C. difficile* colonization. Lastly, repurposing routine MRSA and VRE swabs for *C. difficile* testing allowed for a reasonable period for data collection for a Master's thesis. Therefore, a retrospective cohort study was the ideal study design for this project.

Prospective Cohort Study

Prospective cohort studies involve subject recruitment and exposure data collection before participants developing any outcomes of interest (at baseline) with comparison to non-exposed participants(48). Over time, participants are followed-up through mediums such as mail questionnaires, phone or in-person interviews, physical examinations, or laboratory tests. Disadvantages of a prospective cohort study are that they require long follow-up periods while waiting for outcomes to develop, are vulnerable to high study attrition rates, and can be quite costly. Accompanying the advantages of a prospective cohort study includes increased control an investigator has over data collection. Furthermore, a prospective study design does not deal with recollection bias – which is a common disadvantage of retrospective cohort studies.

In this project's context, however, prospective follow-ups of a patient could become lengthy, especially if they have extended hospital stays and costly as

well – which would not have been suitable for a timeline and budget of a Master's thesis project. A benefit of a prospective study would have been that laboratory samples only needed to be stored for select patients which could have provided relief in terms of storage capacity on the lab. However, adequate data collection in the timeline of a Master's thesis would not have been feasible.

Case-Control Study

Case-control studies involve identifying patients with the outcome of interest at baseline and then reviewing patient history retrospectively to identify potential risk factors(48). A case-control study design consists of a comparison between patients with the outcome of interest and a control. A case-control study is a robust observational study design to determine the importance of a predictor variable in the presence or absence of an outcome of interest.

For the project to be framed as a case-control study, cases and controls would have needed to be matched on variables believed to be confounders, in order to reduce confounding in the sampling design and to increase study efficiency by ensuring cases and controls have similar distributions of confounding variables(49). However, due to limited research on asymptomatic colonization, it is hard to have adequate information on what may be effective confounders of interest to match patients on. Therefore, a retrospective cohort study was the ideal study design.

Study Duration and Enrollment

Participants were considered for study inclusion if they were admitted to general medicine at Hamilton General and The Juravinski Hospitals, or hematology-oncology at The Juravinski Hospital or general medicine and nephrology wards at St Joseph's Healthcare Hamilton between April 2018 and August 2019 (Appendix A). Follow-up information was collected from the date of admission to date of discharge. Based on estimates from the DECEnCY Study Protocol, it was expected that 1000 swabs and accompanying clinical data would be collected from during this time frame with an expected prevalence of colonization of 5-10%.

Study Population

Inclusion Criterion

A patient was included if they fulfilled all of the following criteria: admitted as an in-patient, age > 18 years, MRSA/VRE screening swab collected at either hospital admission or during routine point prevalence screening.

Exclusion Criteria

The pediatric population < 18 years of age were excluded due to differing colonization rates.

STUDY PROCEDURES

MRSA/VRE Screening; Admission & Point Prevalence

The screening protocol for this project is centered around existing Infection Prevention and Control screening policies and procedures for MRSA/VRE at Hamilton Health Sciences and St. Joseph's Hospital Hamilton. Routine universal point prevalence screening occurs monthly on high-risk wards except during outbreaks, where screening occurs weekly. Admission screening for MRSA/VRE occurs for patients considered high-risk and those admitted to the ICU. High-risk MRSA/VRE patients were defined based on prior hospitalization within the previous 12 months and known history of MRSA/VRE colonization. Rectal MRSA/VRE swabs were collected within 72 hours of admission.

Moreover, previous hospital admission has also been found to be a significant risk factor for *C. difficile*, MRSA, and VRE colonization(50). Literature has suggested that a risk-factor screening strategy based on MRSA/VRE risk criteria would identify 74% of all *C. difficile* carriers(43). Therefore, it is plausible that individuals likely to be asymptomatic *C. difficile* carriers are similar to those routinely screened for MRSA/VRE, this assumption forms the basis of the screening strategy for this project.

Swabs for C. difficile Testing

Upon admission or throughout point prevalence screening, nurses collected MRSA/VRE rectal swabs using Copan eSwabs and submitted the swabs in Amies transport medium to the microbiology laboratory as per usual practice. Point prevalence swabs for MRSA/VRE are conducted routinely once per month, and during outbreaks on a weekly basis. Next, swabs of interest based on hospital and ward were sent to the Infectious Diseases Research Laboratory at St. Joseph's Research Institute and screened by LAMP. Positive specimens were verified by PCR, and culture to detect colonization. The LAMP assay targets the *tcd*C gene which is a regulatory element in the gene cassette for *C. difficile* toxin production. PCR for toxin A or binary toxin was used to verify LAMP results. Rectal swabs, as used in the study context for MRSA/VRE surveillance, have close to 100% sensitivity and specificity in *C. difficile* detection, outperforming molecular testing from stool samples, and toxigenic culture of colonized patients(51).

Outcomes: Case Definitions for *C. difficile* Colonization & *C. difficile* Infection

C. difficile Colonization

A case of asymptomatic *C. difficile* colonization was determined by a positive LAMP assay for *tcdC* gene from rectal swab, and confirmed with a positive PCR result for toxin A/binary toxin. A case of symptomatic CDI was determined by a positive LAMP assay from stool of acute diarrhea

Data Sources

Retrospective Chart Review

All patients who met the inclusion criteria, and tested for *C. difficile* colonization had their electronic medical records examined to collect information on potential risk factors of interest, demographic data and presence of *C. difficile* infection as defined by the criteria above.

Abstracted Data Elements from Retrospective Chart Review

Demographic information such as patient age, sex, hospital and ward, date of admission, and discharge were extracted from the computerized hospital information system (MediTech) through MRSA/VRE test sample number. Moreover, patient history about risk factors of interest such as high-risk comorbidities, recent hospitalization, and previous history of *C. difficile* infection was extracted. Length of stay was calculated in the number of days from the date of admission to date of discharge. Lastly, information on symptomatic CDI and mortality was extracted from patient health records as well. The microbiology lab provided data on the results of *C. difficile* colonization.

Data Sample, Sample Size and Power

Based on previously published literature and preliminary results, we anticipated that 5-10% of patients with MRSA/VRE swabs would be positive for colonization of *C. difficile*. Moreover, we expected that we would collect

MSc Thesis – S. George; McMaster University – Health Research Methodology.

approximately 1000 swabs to be tested for *C. difficile* using LAMP and confirmed with PCR over the study period. We also estimated that at least 60-100 carriers of *C. difficile* would be identified. Therefore, assuming the lower end of the prevalence rate of 5%, and with being 95% certain that at least 60 patients would be colonized, a sample size of 1459 patients was needed.

Baseline Characteristics

Demographic and baseline characteristics for the study sample were examined using standard descriptive statistics. For Gaussian continuous variables, averages and standard deviations are reported, and median and interquartile ranges are reported for non-normally distributed data with independent t-tests or Wilcoxon-Mann-Whitney U tests for comparative statistics respectively. Count data and percentages are reported for categorical data, with chi-square tests being used for comparative statistics. A fisher's exact test adjustment was utilized for chi-square tests if at least one cell of a contingency table has n < 5 counts. Baseline admission prevalence rates for asymptomatic colonization were calculated.

<u>ANALYSIS</u>

Primary Analysis – Logistic Regression
A multi-level logistic regression model was conducted to examine possible risk factors for asymptomatic colonization of *C. difficile* at admission (model 1) and point prevalence screening (model 2). The data was represented with two levels consisting of hospital wards and individual patients within hospital wards. The *a priori* hypothesis for needing a hierarchical model is that patients within the same hospital ward may be more similar (clustering), and having hospital ward as the random effect and covariates as fixed effect. A single-level multiple regression model treats the units of analysis as independent observations, therefore resulting in an underestimation of the standard errors of the regression coefficients leading to an overestimation of statistical significance(52).

Both models were generated by first including all the variables of interest: Age, Gender, Previous History of CDI, Recent Hospitalization and Length of Stay (the latter only for model 2). The *a priori* hypothesis was that as subjects increase in age, with a previous history of symptomatic CDI, recent hospitalization in the last 12 months and longer lengths of stay were more likely to have or develop *C*. *difficile* colonization. After examination of the significance of each variable to the overall model (assessed through p values), variables that were not significantly contributing to the model (p-value > 0.10), were removed from the final model, using a backward model building approach. A larger p-value of 0.10 was used as the threshold, as we were interested in a predictive model and a larger p-value is the recommended practice to ensure that important predictors are not

missed(53). Assessment of variable importance from removed variables was computed by comparing the full and reduced models' deviance statistics and AIC values, with lower values of each being indicative of a better performing model. Odds and odds ratios were reported from variables of interest. Adjusted R² were used to assess model performance. The assumptions for the use of logistic regression of a linear relationship between continuous predictor variables and the logit of the outcome variable, and multicollinearity were ruled out. Logistic regression modeling assumes linearity relationship between the independent variables and the log odds, and can be assessed through the examination of scatter plots between each covariate and logit. Logistic regression also requires that the independent variables are not highly correlated with each other and therefore are providing redundant information about the response variable, or little to no multicollinearity. Multicollinearity can be assessed with a variance inflation factor (VIF), and a VIF < 5 is generally tolerated as acceptable. A p value of 0.05 was considered statistically significant.

Moreover, there was the possibility that patients may have repeat admissions within the study time frame, violating the independent observation assumption for logistic regression modelling. If there were very few repeat admissions (<10%), as determined by the author, these data points would count as independent observations. However, if there was a significant proportion of the data with repeat admissions, a sensitivity analysis would be conducted using a

generalized estimating equation (GEE). The GEE's objective is a statistical approach to fit a marginal model that has correlated and clustered responses on the same individual and is a widespread technique commonly used in biomedical research. Repeated ANOVA/MANOVA measures are another method of evaluating correlated or clustered responses. However, it lacks the ability to adjust for covariates – making GEE an appropriate choice. The GEE would be constructed using a logit link function as the dependent variable is binary. Details of the correlation structure and association between the dependent variable and covariates are included in an expression known as the quasi-likelihood function.

Secondary Analysis – Survival Analysis

The exposure of interest was recent hospitalization (within 12 months), age and ward. Demographic information was obtained from electronic health records. Age was grouped as a categorical variable with younger and older than 65.

Exposures of Interest

The exposure of interest was recent hospitalization (within the last12 months), age and ward. Age was grouped as a categorical variable with younger and older than 65, in order to evaluate the effects of being elderly on colonization acquisition.

Ascertainment of Outcomes

The primary endpoint was the development of asymptomatic *C. difficile* colonization. *C. difficile* colonization was defined as via the detection of toxin-specific nucleic acid using laboratory-developed loop-mediated isothermal amplification (LAMP) methods currently used by the Hamilton Regional Laboratory Medicine Program (HRLMP) Virology/Molecular Laboratory.

Covariates of Interest

The previous history of CDI, and sex were determined through a review of electronic health records from in-patient admission.

Statistical Analyses

The study's time origin was admission to any of the study wards. Patients were followed-up from admission and censored at discharge with death as a competing risk. Left censoring was present when a patient was carrier upon admission. A Kaplan-Meier analysis was conducted in order to estimate the probability of survival over time, stratified, as a function of the ward, recent hospitalization, and age differences. A multi-level Cox proportional hazard regression model was used to assess potential risk factors for asymptomatic colonization, and scaled Schoenfeld's residuals were used to assess the proportionality of hazard (PH) assumptions. The proportionality of hazards assumption is that the hazards are proportional over time or the effect of a risk

factor is constant, and is required for the appropriate use of both the log-rank test and Cox PH regression modeling (54). The *a priori* hypothesis for needing a hierarchical model is that patients within the same hospital ward may be more similar, and having hospital ward as the random effect and covariates as fixed effects may help for potential confounding. Cox proportional hazards models were used to estimate the hazard ratio of asymptomatic *C. difficile* colonization. The model was adjusted for covariates of interest including previous colonization, recent hospitalization, age and sex. In a sensitivity analysis, death from non-*C. difficile* related causes were evaluated as a competing risk. If present, patients who were re-admitted to the same hospital ward would be treated as independent observations, and robust standard errors were used to account for intra-patient correlation among repeat admissions. Log-likelihood values will be compared between the null and fitted model to assess model performance. A two-sided pvalue of 0.05 was used to assess statistical significance.

All statistical analyses were performed using RStudio Version 1.1456.

PART III: RESULTS

From January to April 2018, and September 2018 to August 2019, 1470 test swabs were collected from 1056 individual patients and tested for *C. difficile* colonization using LAMP assay and PCR, and used for analysis.

Descriptive Results

From the 1056 patients in the data sample, 72 (6.8%) were identified as asymptomatic *C. difficile* carriers using both admission and point prevalence swabs (Table 1).

The median age at baseline amongst the sample was 72 years old (IQR 61.0-84.0). The median age of colonized patients was 75 years (IQR 61.0-84.0), whereas the non-colonized patient's median age was 72 years (61.0-84.0). Colonized patients were older than non-colonized patients. However, this difference was not statically significant (p = 0.90). From the sample, 535 (50.6%) patients were male. Of the 72 colonized patients, 42 (58.3%) were male, whereas 506/984 (51.8%) non-colonized patients were male. The difference in gender distribution amongst colonized and non-colonized patients was not statistically significant (p = 0.25). The largest proportion of patients were in wards F3 (35.7%) and C3 (19.4%) of The Juravinski Hospital, and Nephrology ward (12.9%) of St Joseph Healthcare Hamilton. The largest proportion of colonized patients are in

wards C3 (19.7%) and F3 (23.4%) from The Juravinski hospital and the Nephrology ward (24.6%) at St. Joseph Healthcare Hamilton. The median length of stay amongst all patients was 19 days (IQR 8.0-41.75). The median length of stay amongst colonized patients was 25 days (IQR 13.0-53.75). The median length of stay amongst non-colonized patients was 19 days (IQR 8.0 – 41.0). The difference between length of stay amongst colonized and non-colonized patients was not statistically significant (p = 0.94).

There were 45 patients with repeat admissions in the sample, resulting in 4.2% of patients having a repeat admission. From all the colonized patients in the sample, 5 (6.9%) had a repeat admission, and 40 (4.06 %) non-colonized patients had a repeated admission. The difference in repeat admission between colonized and non-colonized groups was not statistically significant (p = 0.242). Amongst the sample, 493 (46.6%) patients had been hospitalized within the last 12 months.. Forty-eight of all colonized patients (66.6%) and 445 (45.2%) of all non-colonized patients had a recent hospitalization, with the proportion of recent hospitalization between the two groups being statistically significant (OR 2.41, 95% CI 1.464-4.10).

Among the total sample, 67 patients (6.34%) have had a previous CDI infection. 14 (1.32%) of all colonized patients and 53 (5.01%) of all non-colonized

patients have had a previous case of symptomatic CDI, with the difference in proportion between the two groups being statistically significant (p < 0.001).

Amongst the sample, all but two colonized patients had multiple comorbidities. The most common comorbidities amongst the patients were cardiovascular disease (52%), endocrine including diabetes (41.6%), cancer (26.3%), renal system (26.3%) and gastrointestinal (19.4%). Four patients (5.55%) were missing information on comorbidities from the electronic health records.

Comparison of Testing Strategies

Upon admission, 575 MRSA/VRE swabs were collected from 548 individual patients (Table 2). Amongst the admission swabs, 19 patients were positive for colonization, with two patients having one repeat admission each, resulting in 21 positive admission swabs and a 3.69% colonization rate on admission. Eight hundred ninety-four point prevalence swabs were collected from 615 individual patients, and 51 patients were found to be positive for asymptomatic *C. difficile* colonization, resulting in a prevalence rate of 8.26% for point prevalence testing.

Twelve (16%) patients were negative on admission and later found to be positive for colonization through point prevalence testing. Another twelve patients

had tested negative during point prevalence testing and then were found to be positive during subsequent point prevalence tests; however, they were missing admission testing results. In total, 24/51 patients (33%) had developed *C. difficile* colonization during their hospital stay. Twenty-seven (33.3%) patients had a positive point prevalence test for colonization during hospital admission; however, they were missing test results on admission as well as subsequent point prevalence results throughout their stay; therefore, the natural history of colonization could not be assed amongst this patient group.

When comparing testing methods (admission vs. point prevalence) in identifying asymptomatic *C. difficile* carriers, the difference in prevalence estimates between the two groups is non-significant (OR 0.561, 95% CI 0.016, 7.42). In comparison, those 24 patients who developed *C. difficile* colonization during their hospital stay had a median age of 72.50 years (60.75 - 84.50), whereas those 12 patients who had *C. difficile* colonization on admission had a median age of 70 years (58.75-82.25) (Table 3). Sixteen (72.7%) of the 21 patients positive for *C. difficile* colonization on admission were male, whereas 15 (62.5%) patients were male of those who developed *C. difficile* colonization throughout their hospital stay. Six (28.5%) patients who were positive on admission has a previous case of symptomatic CDI, whereas 1 (4.16%) patient who developed *C. difficile* colonization throughout their hospital stay. Six (28.5%) patients has a previous case of symptomatic CDI, whereas 1 (4.16%) patient who developed *C. difficile* colonization throughout their hospital stay. Six (28.5%) patients has a previous case of symptomatic CDI, whereas 1 (4.16%) patient who developed *C. difficile* colonization throughout their hospital stay. Six (28.5%) patients has a previous case of symptomatic CDI, whereas 1 (4.16%) patient who developed *C. difficile* colonization throughout their hospital stay.

colonization had a recent hospitalization within the last 12 months, 11 (45.8%) patients who developed *C. difficile* during their hospital stay also had a recent hospitalization within the last 12 months. The differences between characteristics were not significant across both groups. There was no significant difference in comorbidities between both groups.

Risk Factors for Colonization, Upon Admission (n = 21)

Risk factors for *C. difficile* colonization at admission are outlined in Table 4. Using a multivariable hierarchical logistic regression model, patients who were colonized on admission (n = 21) had previous CDI (p < 0.001) and female (p< 0.05) as statistically significant for covariates for colonization. Those with previous CDI were 4.76 times more likely to be colonized at admission (OR 4.76, 95% CI 1.49-13.86). Being female was associated with a 2.85 times higher odds of being colonized at admission compared to males (OR 2.66, 95% CI 1.02-8.33).

Assumption Testing and Model Fit

The full model with all predictors (age, sex, previous CDI, recent hospitalization) had an AIC value of 152.8, and Deviance Residuals of 140.8. The R² of the full model was 30.0% indicating moderate to-poor predictive probability. The final model with predictors sex and previous CDI were retained after backwards model selection had an AIC of 150.9 and deviance residual value of 140.9, and an adjusted R² value of 30.15% indicating moderate-to-poor predictive probability. The similarity of the AIC scores indicate that their model fit performances are not well differentiated from each other. The final model performed marginally better than the full model. All assumptions were met in order to perform a logistic regression model.

Risk Factors for Colonization, Point Prevalence (n = 24)

Risk factors for acquisition of C. difficile colonization during hospitalization are outlined in Table 5. Using a multivariable hierarchical logistic regression model, patients who were colonized throughout hospitalization (n = 24) were compared to patients who were not colonized throughout hospitalization (n = n)229). Patients who were not colonized throughout hospitalization included patients who had available negative admission and subsequent point prevalence swabs or, a series of negative point prevalence swabs in the absence of an admission swab. When patients who were colonized throughout a hospital stay were compared to patients who were not colonized throughout their hospital stay. previous CDI (p < 0.001) and recent hospitalization within the last 12 months (p < 0.001) 0.001), were significant predictors. Patients with previous CDI were associated with a 4.77 higher odds to be colonized with C. difficile throughout their hospital stay (OR 4.77, 95% CI 2.15-9.97). Patients with a recent hospitalization were 2.43 times more likely to be colonized with C. difficile than those without a recent hospitalization (OR 2.43, 95% CI 1.29-4.40).

Assumption Testing and Model Fit

The full model with all predictors (Age, sex, previous CDI, recent hospitalization, length of stay) had an AIC value of 398.7, and Deviance Residuals of 384.7. The R² of the full model was 11.86% indicating poor predictive probability. The final model with predictors sex, recent hospitalization and previous CDI after backwards model selection had an AIC of 394.8 and deviance residual value of 384.8, and an adjusted R² value of 12% indicating poor predictive probability. The similarity of the AIC scores indicate that their model fit performances are not well differentiated from each other. The final model performed marginally better than the full model. All assumptions were met in order to conduct a logistic regression model.

Time-to-Colonization of *C. difficile* (n = 24)

Using a multi-level Cox PH model, patients who became colonized during their hospitalization (n = 24) were compared to patients who were not colonized throughout hospitalization (n = 229) (Table 6). Patients missing admission and subsequent point prevalence swabs were excluded from the analysis. The Kaplan-Meier survival curve for time to colonization of asymptomatic *C. difficile* stratified by recent hospitalization demonstrated that patients with a recent hospitalization on average had a greater proportion of patients developing asymptomatic colonization in comparison to those who did not have a recent admission over time. The difference between both survival estimates is

statistically significant, as determined by the log-rank test (p < 0.001). The Kaplan-Meier survival curve for time to the colonization of asymptomatic *C*. *difficile* stratified by Ward demonstrated that the Hemodialysis Outpatient ward (ward 9) stood out with a greater fraction of patients developing asymptomatic colonization over time. The CTU Central Ward (Ward 8) and 8S (Ward 1) had the lowest proportion of patients developing asymptomatic colonization over time. The CTU Central Ward (Ward 8) and 8S (Ward 1) had the lowest proportion of patients developing asymptomatic colonization over time. The difference between rates of asymptomatic colonization by ward was statistically significant (p < 0.001). The Kaplan-Meier survival curve showed that those younger than 65 years had a larger proportion of patients developing asymptomatic colonization over time in comparison to those older than 65 years. As determined by the log-rank test, the difference between the two groups were not statistically significant (p = 0.38).

Using a multilevel Cox proportional hazards model, with ward being the random effect and recent hospitalization, age, being female and previous CDI being the fixed effects, both recent hospitalization (p < 0.001) and previous CDI (p <0.001) were significant predictors. Being recently hospitalized in the past 12 months was associated with a 2 times increase in the hazard of asymptomatic colonization (HR 2.21, 95%CI 1.320,3.731). Having previous CDI was associated with a 2.29 times increase in the hazard of asymptomatic colonization (HR 2.40, 95% CI 1.340,4.305). The proportional hazards assumption was verified using scaled Schoenfeld's residuals, and none of the covariates or global measures

were significant, indicating that the proportionality of hazards assumption was satisfied.

Progression to CDI

From the 72 asymptomatic colonized patients in the sample, 14 went on to develop symptomatic CDI infection (19.44%). From the 21 patients who were positive on admission, 5 (23.8%) developed symptomatic CDI while 16 remained colonized with no symptomatic CDI. From the point prevalence group, 7 went on to develop symptomatic CDI while 17 patients (29.1%) remained colonized with no symptomatic CDI while 17 patients (29.1%) remained colonized with no symptomatic CDI (Figure 1). There were no difference in characteristics between the symptomatic CDI developed from those who were colonized and those who remained colonized with no CDI.

TABLE 1 – Baseline Characteristics of Adult Inpatients Admitted to The Juravinski, Hamilton General or St

Joseph's	Healthcare	Hamilton	Hospitals
----------	------------	----------	-----------

	Overall n = 1056	Colonized n = 72	Not Colonized N = 984	p-value [*] (Comparing Colonized .vs Not Colonized)
Age at baseline in years	72	75	72	0.9
(median, IQR)	(61.0, 84.0)	(61.0,84.0)	(61.0,84.0)	
Sex, Male (%)	535 (50.6)	42 (58.3)	483 (49.0)	0.25
Ward (%)				
Juravinski				
- C3	205 (19.4)	16 (19.7)	189 (19.2)	0.53
- C4	34 (3.21)	1 (1.23)	33 (3.35)	0.36
- E3	66 (6.24)	6 (7.40)	60 (6.09)	0.44
- F3	378 (35.7)	19 (23.4)	359 (36.4)	0.08
Hamilton General				
Hospital	81 (7.66)	2 (2.46)	79 (8.02)	0.10
- 8S	44 (4.16)	5 (6.17)	39 (3.96)	0.22
- 8W				
St Joseph Hamilton				
Healthcare	137 (12.9)	20 (24.6)	117 (11.8)	<0.001 **
- Nephrology	59 (5.58)	0 (0)	59 (5.99)	0.03 **
- CTU(Central)	61 (5.77)	3 (3.70)	58 (5.89)	0.56
- Hemo				
(Outpatient)				
Repeat Admissions (%)	45 (4.26)	5(6.94)	40 (4.06)	0.242
Recent Hospitalization*	493 (46.6)	48 (66.6)	445(45.2)	<0.001**

(within the last 12 months) (%)				
Previous CDI (%)	67 (6.34)	14(1.32)	53 (5.01)	<0.001 **
Length of Stay in days	19 (8.0, 41.75)	25	19	0.94
(median, IQR)		(13, 53.75)	(8, 41.00)	
Comorbidities	-		-	-
- Cardiovascular		52 (72.2)		
- Gastrointestinal		14 (19.4)		
- Endocrine		30 (41.6)		
(including				
Diabetes)				
- Immune		8 (11.1)		
- Cancer		19 (26.3)		
- Neurologic		9 (1.25)		
 Renal System 		19 (26.3)		
 Respiratory 		13 (18.0)		
- Bone		14 (19.4)		
- Other		10 (13.8)		
No Comorbidities		2 (2.77)		
Missing		4 (5.55)		

^{*} For Gaussian continuous variables, averages and standard deviations are reported, and median and interquartile ranges are reported for non-normally distributed data with independent t t-tests or Wilcoxon-Mann-Whitney U tests for comparative statistics respectively. Count data and percentages are reported for categorical data, with chi-square tests being used for comparative statistics

TABLE 2a – Comparison of C. difficile Screening Identification Methods of

High-Risk MRSA/VRE In-patients

	Overall	Positive for	Prevalence
	Overall	Colonization *	Rate
Admission	548	21	3.69%
Swabs			
(n of Patients)			
Point	617	55	8.26%
Prevalence			
Swabs (n of			
Patients)			
Patients (n)	107	12	11.12%
with Both			
Admission and			
Point			
Prevalence			
Swabs			
Patients (n)	171	52	30.4%
with Repeated			
Point			
Prevalence (>			
1) Tests			
Overall	1056	72	6.81%

* As determined through LAMP Assay and confirmed with PCR

.

TABLE 2b – Comparison of C. difficile Screening Identification Methods of

High-Risk MRSA/VRE In-patients

			95% CI	
	n of Patients	OR	(lower,	p-value ***
			upper)	
Patients who				
were	21			
positive on				
admission				
(n = 548)		2.37	(1.29,4.34)	< 0.001
Patients who				
developed				
C. difficile	24			
during				
hospital				
stay**				
(n = 278)				
Patients				
missing				
admission				
and	24	_	_	_
subsequent				
point				
prevalence				
swabs				

*Includes patients who we had evaluable data for. Was calculated using those positive for colonization out of all admission swabs in the numerator. In the denominator, includes those who were followed up and positive for colonization, in comparison to those followed up and negative for admission.

**includes those who had negative admission tests, and had positive point prevalence swabs, and those who had negative point prevalence swabs, and later had positive point prevalence swabs but were missing an initial admission swab

** For gaussian continuous variables, averages and standard deviations are reported, and median and interquartile ranges are reported for non-normally distributed data with independent t t-tests or Wilcoxon-Mann-Whitney U tests for comparative statistics respectively. Count data and percentages are reported for categorical data, with chi-square tests being used for comparative statistics

	Positive for <i>C. difficile</i> at admission n = 21	Negative for <i>C.</i> <i>difficile</i> at admission, positive at point prevalence testing n = 24	<i>p</i> -value *
Age in years	70 (58.75 – 82.25)	72.50 (60.75 – 84.50)	0.40
Sex (% male)	16 (76.1)	15 (65.2)	0.32
Previous CDI (%)	6 (28.5)	1 (4.16)	0.06
Recent Hospitalization (%)	11 (52.3)	11(45.8)	0.77
Comorbidities (%)			
 Cardiovascular Gastrointestinal Endocrine (including Diabetes) Immune Cancer Neurologic Renal System Respiratory Bone Other 	8 (38) 2 (9.5) 11(52.3) 2 (9.5) 3 (14.2) 2 (9.5) 7 (33.3) 3 (14.2) 4 (19.0) 2 (9.5)	15 (62.5) 5 (20.8) 11 (45.8) 2 (8.33) 9 (37.5) 4 (19.0) 10 (41.6) 5 (20.8) 4 (16.6) 2 (8.33)	0.57 0.46 0.38 0.73 0.21 0.66 0.95 0.80 0.62 0.73
No Comorbidities	1 (4.76)	0 (0)	0.23
Missing	3 (14.2)	0 (0)	

TABLE 3 – Comparison of Patients who are Positive at Admission andPositive during Hospital Stay

^{*} For gaussian continuous variables, averages and standard deviations are reported, and median and interquartile ranges are reported for non-normally distributed data with independent t t-tests or Wilcoxon-Mann-Whitney U tests for

comparative statistics respectively. Count data and percentages are reported for categorical data, with chi-square tests being used for comparative statistics

TABLE 4 – P	redictors of C.	difficile Colonization	and Model Fit of	of Multivariable A	Analysis	(Admission)
-------------	-----------------	------------------------	------------------	--------------------	----------	-------------

	Multivariable Analysis (Model 1, Admission, Full Model)			Multivariable Analysis (Model 1, Admission, Final Model)			
	OR	95% CI	<i>p</i> -value	OR	95% CI	<i>p</i> -value	
Age at baseline	1.00	0.97,1.04	0.801	-	-	-	
Sex (Female)	2.85	1.00,9.09	0.06	2.66	1.02,8.33	0.05	
Previous CDI	3.20	0.88,10.29	0.05	4.76	1.49, 13.86	<0.001	
Recent	1.85	0.66,5.67	0.247	-	-	-	
Hospitalization							
Length of Stay	-	-	-	-	-	-	
AIC/Deviance	AIC: 152.8	Deviance: 140.8		AIC: 150.9 Deviance Residuals: 140.9			
Residuals							
R ²	Adjusted R ² = 30.0%			Adjusted $R^2 = 30.15\%$			

TABLE 5 -	- Predictors of (C. difficile	Colonization a	and Model	Fit of Mult	ivariable	Analysis	(Point I	Prevalence)
-----------	-------------------	--------------	-----------------------	-----------	-------------	-----------	----------	----------	-------------

	Multivariable Analysis (Model 1, Admission, Full Model)			Multivariable Analysis (Model 1, Admission, Final Model)			
	OR	95% CI	<i>p</i> -value	OR	95% CI	<i>p</i> -value	
Age at baseline	1.00	0.98-1.02	0.85	-	-	-	
Sex (Female)	1.06	0.59,1.92	0.84	-	-	-	
Previous CDI	4.77	2.15,9.97	<0.001	4.75	2.14,9.94	<0.001	
Recent Hospitalization	2.43	1.29,4.40	<0.001	2.35	1.30,4.42	<0.001	
Length of Stay	0.99	0.99-1.00	0.28				
AIC/Deviance	AIC: 221.6	Deviance: 207.6		AIC: 216.71 Deviance Residuals: 208.7			
Residuals							
	Adjusted R	Adjusted R^2 = 13.94%			Adjusted $R^2 = 14\%$		
R ²	-			-			

TABLE 6 – Results of Cox mixed-effects model on time-to-colonization of

Asymptomatic *C. difficile* Colonization

	ЦВ	95%	CI	n voluo			
	пк	Lower	Upper	p-value			
Age	1.00	0.982	1.019	0.750			
Sex(Female)	1.25	0.77	1.99	0.650			
Recent	2.21	1.320	3.731	0.001 **			
Hospitalization							
Previous CDI	2.40	1.340	4.305	0.001 **			
	Null Log-likelihood: -443.2568						
	Fitted Log-likelihood: -426.016						

	Asymptomatic <i>C.</i> <i>difficile</i> Colonization, with Progression to Symptomatic CDI (n = 14)	Asymptomatic <i>C.</i> <i>difficile</i> Colonization, With no Progression to Symptomatic CDI (n = 58)	OR	95%Cl (lower,upper)	p-values
Age	67.0 (54.0-74.50)	76 (61.25-85.0)	0.95	0.73,1.17	0.61
Sex (% male)	9 (64.2)	37 (63.7)	1.00	0.316,3.412	0.97
Comorbidities (%)					
 Cardiovascular Gastrointestinal Endocrine (including Diabetes) 	4 (28.5) 2 (14.2) 6 (42.8)	48 (82.7) 12 (20.6) 24 (41.3)	0.08 0.64 1.06	0.02,0.32 0.13,3.25 033,3.46	< 0.001 0.58 0.91 0.17
 Immune Cancer Neurologic Renal System Respiratory Bone Other 	3 (21.4) 2 (14.2) 2 (14.2) 3 (21.4) 2 (14.2) 1 (7.1) 1 (7.1)	5 (8.62) 17 (29.3) 7 (12.0) 16 (27.5) 11 (18.96) 13 (22.4) 9 (15.5)	2.89 0.40 1.21 0.72 0.71 0.27 0.42	0.60, 13.92 0.08, 1.99 0.22, 6.60 0.18, 2.90 0.14, 3.65 0.03, 2.23 0.05, 3.61	0.25 0.82 0.63 0.68 0.19 0.41 0.48
No Comorbidities	0 (0.0)	2 (3.44)	-		0.312

TABLE 7 – Comparison of C. difficile Carriage Progression to Symptomatic CDI

Missing	0 (0.0)	4 (6.89)	-		
Previous CDI (%)	5 (35.7)	13 (22.4)	2.51	2.11,2.91	0.30
Recent Hospitalization (%)	11 (78.5)	44 (75.8)	1.10	0.320,4.55	0.83





FIGURE 2 – Kaplan Meier Estimates for Time to Colonization during Hospitalization of Asymptomatic *C. difficile* stratified by Recent Hospitalization



FIGURE 3 – Kaplan Meier Estimates for Time to Colonization during Hospitalization of Asymptomatic *C. difficile* stratified by Ward



FIGURE 4 – Kaplan Meier Estimates for Time to Colonization during Hospitalization of Asymptomatic *C. difficile* stratified by Age Categories younger and older than 65 years



Recent Hospitalization 🕂 < 65 Years 🕂 65 + Years

PART IV: DISCUSSION

Summary of Findings

Our retrospective cohort study included 1056 individual patients admitted to 9 inpatient wards over 15 months, with 72 positive LAMP tests for asymptomatic colonization, resulting in an overall prevalence rate of 6.81% (range 0% - 24.6%). There were no cases of asymptomatic colonization on the CTU (Central) ward in SJHH. The Nephrology ward in SJHH had the largest percentage (24.6%) of patients who were asymptomatically colonized throughout the study period. Twenty-one (3.69%) of patients with admission swabs were colonized with C. *difficile*, and 24 (8.63%) patients became colonized during their hospital stay. Twenty-four patients had positive point prevalence swabs, however with missing admission swabs or subsequent point prevalence swabs, therefore establishing the natural history of colonization amongst this group was not possible. In comparing testing strategies, there were no significant differences in the identification of inpatients; however, point prevalence testing identified an additional 51 cases and a testing strategy that involves both point prevalence, and admission screening is imperative. Risk factors for colonization at admission were and previous CDI, while risk factors for colonization at point prevalence were previous CDI and recent hospitalization. Those with a recent hospitalization and from Hemodialysis and Nephrology wards had the most significant proportion of patients who progressed to asymptomatic colonization over time. Recent

Hospitalization and previous CDI were also significant predictors in the hazard rate of *C. difficile* colonization. Fourteen colonized patients developed symptomatic CDI; however, there were not significant differences in sex, age, previous CDI, previous hospitalization and comorbidities, except for cardiovascular diseases between those who developed symptomatic CDI and those who remained asymptomatic and colonized.

Prevalence of Asymptomatic C. difficile Colonization

Over the collection period of January to April 2018 and September 2018 to April 2019, *C.*

difficile colonization prevalence across a mixed adult inpatient population was 6.81%. Compared to other studies identified in the narrative review, the colonization rate of 6.81% falls within the lower end of the reported patient colonization rate of 4-29%(11, 36). Higher estimates of colonization rates in studies have been due to differences in a patient population, study location, testing strategies and laboratory methods. The prevalence rate on admission was 3.69%, consistent with reported admission colonization rates of other published literature. For example, general internal medicine or mixed wards with adult inpatients in the United States, reported prevalence rates for asymptomatic carriage are in the range of 4.4-15.4%. Two larger-scale studies by Kong et al.

and Longtin et al. had a prevalence rate of 4.05% and 4.8%, respectively(36, 42). However, it is relevant to note that both of the studies, as mentioned earlier, had a larger sampled population size (7000-9000 sampled patients) and may suggest why colonization rates for these studies were slightly higher as our study did not meet sample size recruitment necessary to detect our desired prevalence rate. The prevalence rate for colonization amongst point prevalence testing strategy was 8.26%. Studies of prevalence rates of hospitalized patients have been shown to range from 3-21%. Literature has suggested a 50% acquisition rate for patients with a length of stay greater than four weeks for asymptomatic *C. difficile* colonization. The failure to reach adequate sample size by 403 patients may indicate why our prevalence estimate was on the lower end of the scale.

Twenty-one patients were identified on admission as positive for colonization, while point prevalence swabs identified 24 patients who developed colonization throughout their hospital stay, identifying an additional 54% of carriers. Therefore, a joint strategy of identifying patients on admission in addition to point prevalence screening is ideal. This finding is consistent with the literature. One study had found an admission prevalence rate of 9.5%, and weekly testing revealed an additional 7.2% of patients that acquired carriage after admission. Another study found an increase of 50% in colonization rates over a hospital stay despite a low (2.1%) admission rate. The difference between in-patients screened on admission and those with point-prevalence screenings were

statistically significant (p = 0.001) in carriage identification, with universal point prevalence screening identifying 2.37 times more colonization patients than admission screening along (OR 2.37, 95% CI 1.29, 4.34) suggesting that the addition of universal point prevalence screening in addition to admission screening helped identify more than double the amount of carriers in the population.

Risk Factors for Asymptomatic Colonization, at admission and point prevalence

Our study identified that being female and previous CDI were risk factors for *C*. *difficile* colonization on admission. Recent hospitalization and previous CDI were risk factors for *C*. *difficile* colonization throughout a hospital stay. These risk factors are consistent with our findings from the narrative review for the majority. One study had found that previous CDI and recent hospital stay were significant predictors for colonization on admission, and the use of proton pump inhibitors, corticosteroids, and antibiotic use before admission were not significant predictors. Our study agrees with these results; however, data on medication history was not collected and suggest future research steps and a limitation of the study.

MSc Thesis – S. George; McMaster University – Health Research Methodology.

Moreover, literature has also suggested that risk factors for developing *C. difficile* during hospital stay were previous hospitalization, proton pump inhibitors, H2 blockers, chemotherapy, previous CDI, an increased number of comorbidities. Our findings are somewhat consistent with the literature, as we identified previous hospitalization and previous CDI as risk factors of importance. Still, as previously mentioned, an exploration of medication therapy and comorbidities were not analyzed in this study. Comorbidities amongst patients, especially those known to be high risks such as cancer and gastrointestinal diseases and medication therapy would be useful for further research, and could improve the predictive probability of the regression model. The large range of 95% confidence intervals from parameter estimates are indicative of the analysis being underpowered, due to an inadequate sample size.

Our study sample had a repeat admission of less than 10%; therefore, a GEE correction was not used.

Time-to-Colonization

The Kaplan-Meier estimates revealed that patients with a recent hospitalization, and being admitted to hematology and nephrology wards were at higher risk for developing asymptomatic colonization over time. This is consistent with the literature, as recent admission is consistently an identified risk factors for *C*.

difficile colonization development. Moreover, studies have identified comorbidities of high risk for *C. difficile* colonization to include renal-related comorbidities and immunosuppressed patients, such as those diagnosed with hematologic cancers, which are typical patients on both Nephrology and Hematology wards(47). Moreover, the multilevel cox proportional hazard model identified previous CDI and recent hospitalization to be risk factors for developing *C. difficile* colonization throughout a hospital admission. These results reiterate what has been found in the logistic regression models and solidify what had been identified in the literature.

Progression to CDI

The role of *C. difficile* colonization and the development of symptomatic CDI is not well understood. 19.4% (14/72) of colonized patients in the study went on to develop symptomatic CDI. Our prevalence of progression to symptomatic CDI is consistent with the literature. In a published systematic review, 21.8% (95% CI 7.9-40.1%) of colonized adult inpatients on admission developed a subsequent CDI episode during their hospital stay. However, another study reported rates as low as 2.5% in progression to symptomatic CDI, a mixed lower-risk patient population. The significant heterogeneity in the progression results to symptomatic CDI from asymptomatic colonization can vary from reason such as strain, testing methods, and other

patient population differences. Age, sex, previous CDI, recent hospitalization, and comorbidities except for cardiovascular diseases (OR 0.08, 95% CI 0.02-0.32) were not risk factors in the development of symptomatic CDI amongst those already asymptomatically colonized, with cardiovascular disease being protective of symptomatic CDI development.

Limitations

One of the limitations of this study was the processes in data collection. In the beginning stages, swabs were supposed to be marked with a sticker by the nurse on the ward, and when received at the microbiology lab forwarded to the research laboratory for storage until testing may need to be required. However, some of the labels were not sticking to the physical tubes of samples, and often, the nursing staff forgot to attach stickers. Furthermore, stickered specimens may not have been reliable put aside by the laboratory personnel. Thus swabs were getting thrown away as they were thought to not be from wards of interest and resulted in thinking that there was an under-collection of swabs from wards of interest. It wasn't for several months until the lab-processed could get changed to have all swabs reliably collected, and individually screened by hand to make sure they were from wards of interest. This new strategy was more labour intensive but ensured that no specimens went missing. However, the lack of a reliable process prior to implementation of this new process resulted in missing
specimens and in a segmented collection date with no swabs collected between April 2018 and August 2018. Moreover, due to the global pandemic of COVID-19 being announced in early March, laboratory operations and *C. difficile* colonization testing were not available to add more samples to the patient population, whether it was more colonized inpatients or more point prevalence swabs of already admitted patients. Therefore, the sample size was not met by 403 patients. Moreover, not all patients had available admission and subsequent point prevalence swabs with only 24 out of 51 point prevalence swabs available for evaluation. The limitations in not achieving sample size, may result in models being statistically underpowered to detect a significant difference.

Another limitation of the dataset is due to secondary data review. Because there was no primary data collection in the study, information was extracted from electronic health records. However, there was inconsistent information on variables such as admission and discharge date, and clarity in medication therapies and current comorbidities, which would have been useful information for the nature of this study.

In addition to the limitations, as mentioned above, predictors were chosen based on the literature. Age and gender were accurately captured, but if CDI occurred at another facility outside of HHS; there was no information on the occurrence, the same for recent hospitalization. Moreover, recent hospitalization

61

was defined as a hospitalization within 12 months prior to admission, and therefore the impact of the duration (i.e. long vs short hospital stay) of the previous hospitalization was not assessed.

PART V: CONCLUSION & FUTURE STEPS

In conclusion, the results of this study add significantly to the existing literature currently on C. difficile colonization. The overall admission and point prevalence colonization prevalence rates add to the current estimates of C. difficile colonization. Moreover, the review of different testing strategies show that a mixed procedure involving both admission and point prevalence testing is essential in order to identify the majority of patients colonized with C. difficile. Our risk factors identified on admission for C. difficile colonization are being female and having recent hospitalization. Our risk factors identified during hospitalization for colonization were previous CDI and recent hospitalization. Our survival analysis model that demonstrated that high-risk wards including hemodialysis and nephrology can inform health care providers to which patients may be high risk to be carriers for C. difficile colonization. Lastly, our study on progression to CDI identified that approximately 20% of colonized patients would progress to symptomatic CDI throughout their hospital stay which can be informative for decision making in the future.

Future steps for this research would be to collect data on comorbidities and antibiotic therapy to assess these variables in the natural history of *C. difficile* colonization. Moreover, this study was conducted in three acute care sites across Hamilton, Ontario. A more diverse study sample including multiple cities would

63

MSc Thesis – S. George; McMaster University – Health Research Methodology.

help make the results of this study more applicable across Ontario. Due to issues in consistent data collection over a period of time, a future step would be to obtain data from a more consistent frame in order to study potential seasonality of colonization acquisition. Lastly, in order to truly assess the role of asymptomatic *C. difficile* carriers in the transmission and contribution to hospitalwide outbreaks, epidemiologic links between cases should be established. Whole-genome sequencing and contact tracing methods in order to determine what proportion of those with symptomatic CDI develop disease resulting from their colonizing strain acquired before admission to the hospital would be a direction of further investigation.

At this time, our study adds to the existing literature testing strategies and risk factors for *C. difficile* colonization, and progression to CDI amongst colonized patients.

Site	Colour	Ward	Number
Hamilton General Hospital (HGH)	Pink	8S	1
		8W	2
Juravinski Hospital	Yellow	C3	3
		C4	4
		E3	5
		F3	6
St. Joseph's Healthcare Hamilton (SJHH)	Green	Nephrology	7
		CTU (Central)	8
		Hemodialysis Outpatients	9

APPENDIX

REFERENCE

1. Curry SR. Clostridium difficile. Clin Lab Med. 2010;30(1):329-42.

2. DePestel DD, Aronoff DM. Epidemiology of *Clostridium difficile* Infection. J Pharm Pract. 2013;26(5):464–75.

3. Katz KC, Golding GR, Choi KB, Pelude L, Amaratunga KR, Taljaard M, et al. The evolving epidemiology of *Clostridium difficile* infection in Canadian hospitals during a postepidemic period (2009–2015). CMAJ; 2018. p. E758-65.

4. Canada Go. Fact Sheet - Clostridium difficile (C. difficile) 2014 [Available from: <u>https://www.canada.ca/en/public-health/services/infectious-diseases/fact-sheet-clostridium-difficile-difficile.html</u>.

5. Mada PK, Alam MU. Clostridium Difficile. StatPearls Publishing; 2020.

6. Rineh A, Kelso MJ, Vatansever F, Tegos GP, Hamblin MR. Clostridium difficile infection: molecular pathogenesis and novel therapeutics. 2014;12(1):131-50.

Heinlen L, Ballard JD. *Clostridium difficile* Infection. Am J Med
 Sci 2010;340(3):247–52.

8. Bella SD, Ascenzi P, Siarakas S, Petrosillo N, Masi Ad. *Clostridium difficile* Toxins A and B: Insights into Pathogenic Properties and Extraintestinal Effects. CMAJ. 2018;190(25):E758-E65.

9. Hookman P, Barkin JS. *Clostridium difficile* associated infection, diarrhea and colitis. World J Gastroenterol. 2009;15(13):1554–80.

10. Awad MM, Johanesen PA, Carter GP, Rose E, Lyras D. Clostridium difficilevirulence factors: Insights into an anaerobic spore-forming pathogen.

11. Luis Furuya-Kanamori, John Marquess, Laith Yakob, V. Riley T,

L. Paterson D, F. Foster N, et al. Asymptomatic Clostridium difficile colonization: epidemiology and clinical implications. BMC Infectious Diseases. 2015;15(1).

12. Klaas Smits W, Dena Lyras, Borden Lacy D, H. Wilcox M, J. Kuijper E. Clostridium difficile infection. Nature Reviews Disease Primers. 2016;2(1):16020.

13. Ontario Agency for Health Protection and Promotion PIDACP. Annex A:

Screening, Testing and Surveillance for Antibiotic-Resistant Organisms (AROs) In All Health Care Settings. Toronto, ON: Queen's Printer for Ontario,; 2013.

14. McDonald LC, Lessa F, Sievert D, Wise M, Herrera R, Gould C, et al. Vital Signs: Preventing *Clostridium difficile* Infections. Morbidity and Mortality Weekly Report (*MMWR*). 2012;61(09):**157-62**.

15. Shankar Kumar, Ster Irina C, Richard Pollok, Ivan Muscat, D. Planche T. The Epidemiology of Community Clostridium difficile Infection: A Five-Year Population-Based Study on the Bailiwick of Jersey, Channel Islands. Infection Control & Hospital Epidemiology. 2018;39(5):603-7.

16. C. Lessa F, Yi Mu, M. Bamberg W, G. Beldavs Z, K. Dumyati G, R. Dunn J, et al. Burden of Clostridium difficileInfection in the United States. New England Journal of Medicine. 2015;372(9):825-34.

S. Chitnis A, M. Holzbauer S, M. Belflower R, G. Winston L, M. Bamberg
 W, Carol Lyons, et al. Epidemiology of Community-AssociatedClostridium
 difficileInfection, 2009 Through 2011. JAMA Internal Medicine.

2013;173(14):1359.

 Sahil Khanna, Arjun Gupta. Community-acquired Clostridium difficile infection: an increasing public health threat. Infection and Drug Resistance.
 2014:63.

19. L Kuntz J, A Chrischilles E, F Pendergast J, A Herwaldt L, M Polgreen P. Incidence of and risk factors for community-associated Clostridium difficile infection: A nested case-control study. BMC Infectious Diseases. 2011;11(1):194.

20. Cole SA, Stahl TJ. Persistent and Recurrent Clostridium difficile Colitis. Clinics in Colon and Rectal Surgery. 2015;28(2):65-9.

21. Rawish F, Muhammad A. The Hypervirulent Strain of Clostridium Difficile: NAP1/B1/027 - A Brief Overview. Cureus; 2019. p. 3977.

22. Public Health Agency of Canada GoC. Clostridium Difficile Infection 2013 [Available from: https://www.canada.ca/en/public-health/services/infectiousdiseases/nosocomial-occupational-infections/clostridium-difficile-infectionprevention-control-guidance-management-long-term-care-facilities.html.

A. Pereira J, Allison Mcgeer, Antigona Tomovici, Alex Selmani,
 Ayman Chit. Incidence and economic burden of Clostridioides difficile infection in
 Ontario: a retrospective population-based study. CMAJ Open. 2020;8(1):E16 E25.

24. J. Freeman, P. Bauer M, D. Baines S, J. Corver, N. Fawley W,

B. Goorhuis, et al. The Changing Epidemiology of Clostridium difficile Infections. Clinical Microbiology Reviews. 2010;23(3):529-49.

25. Sahil Khanna, S Pardi D, L Aronson S, P Kammer P, Robert Orenstein, Sauver JL S, et al. The Epidemiology of Community-Acquired Clostridium difficile Infection: A Population-Based Study. American Journal of Gastroenterology. 2012;107(1):89-95.

26. R. Reveles K, V. Pugh MJ, A. Lawson K, M. Mortensen E, M. Koeller J, R. Argamany J, et al. Shift to community-onset Clostridium difficile infection in the national Veterans Health Administration, 2003-2014. American Journal of Infection Control. 2018;46(4):431-5.

27. I. See, Y. Mu, J. Cohen, G. Beldavs Z, G. Winston L, G. Dumyati, et al. NAP1 Strain Type Predicts Outcomes From Clostridium difficile Infection. Clinical Infectious Diseases. 2014;58(10):1394-400.

28. J. Forster A, M. Taljaard, N. Oake, K. Wilson, V. Roth, Walraven CV. The effect of hospital-acquired infection with Clostridium difficile on length of stay in hospital. Canadian Medical Association Journal. 2012;184(1):37-42.

 Natasha Nanwa, Beate Sander, Murray Krahn, Nick Daneman, Hong Lu,
 C. Austin P, et al. A population-based matched cohort study examining the mortality and costs of patients with community-onset Clostridium difficile infection identified using emergency department visits and hospital admissions. PLOS ONE. 2017;12(3):e0172410. 30. R. Levy A, M. Szabo S, Greta Lozano-Ortega, Elisa Lloyd-Smith,
Victor Leung, Robin Lawrence, et al. Incidence and Costs of Clostridium difficile
Infections in Canada. Open Forum Infectious Diseases. 2015;2(3):ofv076.

R. Curry S, A. Muto C, L. Schlackman J, William Pasculle A, A. Shutt K,
W. Marsh J, et al. Use of Multilocus Variable Number of Tandem Repeats
Analysis Genotyping to Determine the Role of Asymptomatic Carriers in
Clostridium difficile Transmission. Clinical Infectious Diseases. 2013;57(8):1094102.

32. Clabots CR, Johnson S, Olson MM, Peterson LR, Gerding DN. Acquisition of Clostridium difficile by hospitalized patients: evidence for colonized new admissions as a source of infection. J Infect Dis. 1992;166(3):561-7.

33. Yuan-Pin Hung, Pei-Jane Tsai, Kuei-Hsiang Hung, Hsiu-Chuan Liu, Chih-I Lee, Hsiao-Ju Lin, et al. Impact of Toxigenic Clostridium difficile Colonization and Infection among Hospitalized Adults at a District Hospital in Southern Taiwan. PLoS ONE. 2012;7(8):e42415.

M. Riggs M, K. Sethi A, F. Zabarsky T, C. Eckstein E, P. Jump RL,
J. Donskey C. Asymptomatic Carriers Are a Potential Source for Transmission of Epidemic and Nonepidemic Clostridium difficile Strains among Long-Term Care Facility Residents. Clinical Infectious Diseases. 2007;45(8):992-8.

35. Armin S, Shamsian S, Drakhshanfar H. Colonization with Clostridium difficile in Children with Cancer. Iranian Journal of Pediatrics. 2013;23(4):473-6.

Yuan Kong L, Nandini Dendukuri, Ian Schiller, Anne-Marie Bourgault,
 Paul Brassard, Louise Poirier, et al. Predictors of asymptomatic Clostridium
 difficile colonization on hospital admission. American Journal of Infection Control.
 2015;43(3):248-53.

37. Lyudmila Boyanova, Nikolay Kalvatchev, Daniel Yordanov,

Petyo Hadzhiyski, Rumyana Markovska, Galina Gergova, et al. Clostridioides (Clostridium) difficile carriage in asymptomatic children since 2010: a narrative review. Biotechnology & Biotechnological Equipment. 2019;33(1):1228-36.

38. Tenover FC, Baron EJ, Peterson LR, Persing DH. Laboratory Diagnosis of *Clostridium difficile* Infection Can Molecular Amplification Methods Move Us Out of Uncertainty? . J Mol Diagn. 2011;13(6):573-82.

39. Mohan SS, McDermott BP, Parchuri S, Cunha BA. Lack of value of repeat stool testing for Clostridium difficile toxin. Am J Med. 2006;119(4):356.

40. L. Boyanton B, P. Sural, R. Loomis C, C. Pesta, L. Gonzalez-Krellwitz,
B. Robinson-Dunn, et al. Loop-Mediated Isothermal Amplification Compared to
Real-Time PCR and Enzyme Immunoassay for Toxigenic Clostridium difficile
Detection. Journal of Clinical Microbiology. 2012;50(3):640-5.

41. Kim H, Riley TV, Kim M, Kim CK, Yong D, Lee K, et al. Increasing Prevalence of Toxin A-Negative, Toxin B-Positive Isolates of Clostridium difficile in Korea: Impact on Laboratory Diagnosis. Journal of Clinical Microbiology. 2008;46(3):1116-7.

42. Yves Longtin, Bianka Paquet-Bolduc, Rodica Gilca, Christophe Garenc, Elise Fortin, Jean Longtin, et al. Effect of Detecting and IsolatingClostridium difficileCarriers at Hospital Admission on the Incidence of CdifficileInfections. JAMA Internal Medicine. 2016;176(6):796.

43. Leekha S, Aronhalt KC, Sloan LM, Patel R, Orenstein R. Asymptomatic Clostridium difficile colonization in a tertiary care hospital: admission prevalence and risk factors. Am J Infec Control. 2013;41(5):390-3.

44. Holger Schäffler, Anne Breitrück. Clostridium difficile – From Colonization to Infection. Frontiers in Microbiology. 2018;9.

45. M Zacharioudakis I, N Zervou F, Eleftheria Pliakos E, D Ziakas P, Eleftherios Mylonakis. Colonization With Toxinogenic C. difficile Upon Hospital Admission, and Risk of Infection: A Systematic Review and Meta-Analysis. American Journal of Gastroenterology. 2015;110(3):381-90.

46. Alonso CD, Treadway SB, Hanna DB, Huff CA, Neofytos D, Carroll KC, et al. Epidemiology and Outcomes of Clostridium difficile Infections in

Hematopoietic Stem Cell Transplant Recipients. Clinical Infectious Diseases. 2012;54(8):1053-63.

47. T. Crobach MJ, J. Vernon J, G. Loo V, Yuan Kong L, Séverine Péchiné,
H. Wilcox M, et al. Understanding Clostridium difficile Colonization. Clinical
Microbiology Reviews. 2018;31(2).

48. W. Song J, C. Chung K. Observational Studies: Cohort and Case-Control Studies. Plastic and Reconstructive Surgery. 2010;126(6):2234-42.

49. Sherri Rose, Laan MJVD. Why Match? Investigating Matched Case-Control Study Designs with Causal Effect Estimation. The International Journal of Biostatistics. 2009;5(1).

50. PIDAC. Annex A: Screening, Testing and Surveillance for Antibiotic-Resistant Organisms (AROs).

51. Curry S, Schlackman J, Hamilton T. Perirectal swab surveillance for Clostridium difficile by use of selective broth preamplification and real-time PCR detection of tcdB. J Clin Microbiol. 2011;49(11):3788-93.

52. Modelling UoBCfM. What are multilevel models and why should I use them? [Available from: <u>http://www.bristol.ac.uk/cmm/learning/multilevel-models/what-why.html</u>.

53. Maatheus M. Model Building. Swiss Federal Institute of Technology Zurich; 2008.

54. Wayne W. LaMorte M, PhD, MPH. Survival Analysis: Boston University School of Public Health; 2016 [Available from:

https://sphweb.bumc.bu.edu/otlt/mph-

modules/bs/bs704 survival/BS704 Survival7.html.